Author’s response to reviews

Title: An unusual case of acute lupus haemophagocytic syndrome: a test of diagnostic criteria

Authors:

Udai Wijetunga (udaiwijetunga@gmail.com)
Ravindra Satarasinghe (ravindrasatharasinghe@yahoo.com)
Balasuriya Dayananda (dayananda2000@yahoo.com)
Ganhewage Darshani (darshikokila@yahoo.com)

Version: 3 Date: 12 Apr 2017

Author’s response to reviews:

1. Page 2; Line 42: Add "The" before "Usual patient" Should read, "The usual patient presentation of HLH is characterized by prolonged fever..."

Response:

Revision done.

2. P3L39: Why does intravenous begin with a capital?

Response:

It was a typographical error. Revision done.

3. P5L41: Diagnostic criteria, not diagnosis criteria

Response:

Revision done. The same error was corrected elsewhere as well.

4. P6L50: Add "The" before "occurrence of ALHS is usually seen in the setting of high ANA levels..."

Response:
Revision done.